Author's response to reviews

Title: Delivery Practices and Associated Factors among Mothers seeking child welfare services in Nyandarua South District, Kenya

Authors:

Carol Wanjira (cwkagia@gmail.com)
Moses Mwangi (mmwangi@kemri_nuitm.or.ke)
Evans Mathenge (emathenge@domckeny.or.ke)
Gabriel Mbugua (mutitu2002@yahoo.ca)
Zipporah Ng'ang'a (zipnganga@gmail.com)

Version: 3 Date: 20 December 2010

Author's response to reviews: see over
To the Editor,

Kindly find attached the manuscript submission for the “Delivery Practices and Associated Factors among Mothers seeking child welfare services in Nyandarua South District, Kenya”.

I believe that this work is suitable for publishing in the BMC Public health as it touches on maternal health, one of the major aspects of public health importance in Kenya. Though there have been studies carried out on the topic of factors affecting mothers choice of delivery place, however these factors vary from place to place thus site specific studies are important. In the study described in the manuscript, we have covered a unique perspective on maternal health in one of the most populous districts in Kenya. We have explored mothers’ perceptions on birth attendants, a factor that has not been investigated to the best of our knowledge.

Thank you.